Author's response to reviews

Title: Implementing Comprehensive PMTCT and HIV Prevention for South African Couples: study protocol for a randomized clinical trial

Authors:

Deborah Jones (d.jones3@med.miami.edu)
Karl Peltzer (karl.pel@mahidol.ac.th)
Stephen M. Weiss (SWeiss2@med.miami.edu)
Sibusiso Sifunda (SSifunda@hsrc.ac.za)
Ntabozuko Dwane (ndwane@hsrc.ac.za)
Shandir Ramlagan (sramlagan@hsrc.ac.za)
Ryan Cook (rcook@med.miami.edu)
Gladys Matseke (gmatseke@hsrc.ac.za)
Vincent Maduna (vmaduna@hsrc.ac.za)
Andrew Spence (spencea@med.miami.edu)

Version: 3 Date: 10 June 2014

Author's response to reviews:

Below is done

Please ensure the title conforms to journal style for study protocol articles. The title should follow the format “________________: study protocol for a randomized controlled trial.”

2. Please include the date of registration with your trial registration number at the end of your Abstract.

3. As your research involves humans please include a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number. Any experimental research on humans must be in compliance with the Helsinki Declaration. If your study is taking place in multiple sites please ensure you mention all of the ethical bodies that have approved the study at the different sites.

4. Please include a list of abbreviations used and their meanings, after your Trial Status.

5. Please include a figure title and legend section after your reference list.